Author's response to reviews

Title: Decentralization of the provision of health services to people living with HIV/AIDS in rural China: the case of three counties

Authors:

Xiulan Zhang (zhang99@bnu.edu.cn)
Pierre Miège (miege@hotmail.com)
Yurong Zhang (yurongbnu@gmail.com)

Version: 6 Date: 28 January 2011

Author's response to reviews:

Dear Editor,

changes have been done according to your requirements. Especially, a small mistake was found in the numbering of footnotes within the text (note 19 had been mistakenly replaced by note 16).

Thank you for your attention, sincerely,

Pierre Miege